# **ETH** zürich

# Observed Trends in Global Indicators of Mean and Extreme Streamflow

**Journal Article** 

Author(s): <u>Gudmundsson, Lukas</u> (); Leonard, Michael; Do, Hong X.; Westra, Seth; <u>Seneviratne, Sonia I.</u>

Publication date: 2019-01-28

Permanent link: https://doi.org/10.3929/ethz-b-000328047

Rights / license: Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International

**Originally published in:** Geophysical Research Letters 46(2), <u>https://doi.org/10.1029/2018GL079725</u>

**Funding acknowledgement:** 617518 - Land-Climate Interactions: Constraints for Droughts and Heatwaves in a Changing Climate (EC)

# **Geophysical Research Letters**

# **RESEARCH LETTER**

10.1029/2018GL079725

#### **Key Points:**

- A global assessment of trends in streamflow covering 1951–2010 and 14 subcontinental regions
- The significance of regional trends is assessed, revealing complex spatiotemporal change patterns
- Indicators of mean and extreme (low and high) streamflow often share the same sign of change

Supporting Information:

Supporting Information S1

#### Correspondence to:

L. Gudmundsson, lukas.gudmundsson@env.ethz.ch

#### **Citation:**

Gudmundsson, L., Leonard, M., Do, H. X., Westra, S., & Seneviratne, S. I. (2019). Observed trends in global indicators of mean and extreme streamflow. *Geophysical Research Letters*, 46, 756–766. https://doi.org/ 10.1029/2018GL079725

Received 22 JUL 2018 Accepted 20 DEC 2018 Accepted article online 26 DEC 2018 Published online 23 JAN 2019

#### ©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# **Observed Trends in Global Indicators** of Mean and Extreme Streamflow



<sup>1</sup>Institute for Atmospheric and Climate Science, Department of Environmental Systems Science, ETH, Zurich, Switzerland, <sup>2</sup>School of Civil, Environmental and Mining Engineering, University of Adelaide, Adelaide, South Australia, Australia, <sup>3</sup>Faculty of Environment and Natural Resources, Nong Lam University, Ho Chi Minh City, Vietnam

**Abstract** This study investigates global changes in indicators of mean and extreme streamflow. The assessment is based on the Global Streamflow Indices and Metadata archive and focuses on time series of the annual minimum, the 10th, 50th, and 90th percentiles, the annual mean, and the annual maximum of daily streamflow. Trends are estimated using the Sen-Theil slope, and the significance of mean regional trends is established through bootstrapping. Changes in the indices are often regionally consistent, showing that the entire flow distribution is moving either upward or downward. In addition, the analysis confirms the complex nature of hydrological change where drying in some regions (e.g., in the Mediterranean) is contrasted by wetting in other regions (e.g., North Asia). Observed changes are discussed in the context of previous results and with respect to model estimates of the impacts of anthropogenic climate change and human water management.

**Plain Language Summary** Studies of trends in streamflow data from across the globe are essential for understanding patterns and changes in water availability (e.g., regions of deficit and abundance) and evaluating the fidelity of global water availability models. This study evaluates historical trends in streamflow data, using a new data set of observations from over 30,000 sites around the world. The study is comprehensive, looking at changes in low flows (defined as the lowest day of flow in each year), average flows, and high flows (the highest day of flow in each year). An interesting outcome is that where trends are present in a region, the direction of the trend is often consistent across all indices for that region (consistently drier or wetter), as distinct from the possibility of stronger extremes (wetter maximums and drier minimums).

#### 1. Introduction

Among the most important implications of anthropogenic climate change are the potential for both largescale changes in water availability (Greve et al., 2018; Schewe et al., 2014) and increases in the magnitude and occurrence of floods and droughts (Hirabayashi et al., 2013; Prudhomme et al., 2014). Simultaneously, the unprecedented scale of on-ground human interventions in the water cycle—including reservoir construction, irrigation, and land cover change—is also affecting terrestrial hydrology and might even exceed the impact of future climate change in some regions (Haddeland et al., 2014).

To better anticipate future changes in the world's water resources and hydrological extremes, it is essential to analyze already observed changes. Among all components of the terrestrial water cycle, streamflow (including river flow) is arguably the variable that has been monitored with the highest station density and the longest temporal coverage (Fekete et al., 2012, 2015; Hannah et al., 2011) and is thus the best we have for investigating past changes in water resources and hydrological extremes.

An increasing number of regional studies have drawn a complex picture of trends in annual streamflow statistics over several (sub)continents, including North America (Burn & Elnur, 2002; Douglas et al., 2000; Hodgkins et al., 2017; Lettenmaier et al., 1994; Lins & Slack, 1999; Mallakpour & Villarini, 2015; McCabe & Wolock, 2002; McClelland et al., 2006), South America (Genta et al., 1998; Marengo et al., 1998; Pasquini & Depetris, 2007), Europe (Blöschl et al., 2017; Gudmundsson et al., 2017; Hannaford et al., 2013; Hisdal et al., 2001; Hodgkins et al., 2017; Stahl et al., 2010), and Asia (Adam & Lettenmaier, 2008; MacDonald et al., 2007; McClelland et al., 2006; Tananaev et al., 2016). It is, however, difficult to generalize from these assessments, as they are often tailored to match conditions in a specific continent, consider





different periods, and have variations in method and selected indices. Furthermore, little work has been published for several important landmasses, including Africa and large parts of Eurasia.

Of the global studies that focus on trends in observed streamflow, some are dedicated to changes in the total freshwater fluxes to the ocean, thereby focusing on the outlets of continental-scale river basins (Alkama et al., 2011; Dai et al., 2009; Dai & Trenberth, 2002; Labat et al., 2004; Milliman et al., 2008). Consistent with the regional studies, these highlight spatially complex trend patterns. Although these assessments are of high relevance for ocean and Earth system dynamics, they focus on the net terrestrial water balance and cannot infer regional- to local-scale changes.

Another branch of global studies has assessed streamflow trends of individual water bodies. Some studies have focused on investigating changes in a few carefully selected large river basins (Jaramillo & Destouni, 2015; Kundzewicz et al., 2005; Milly et al., 2005; Svensson et al., 2005), thereby taking advantage of better quality control of the individual records but suffering from relatively small sample sizes and sparse spatial coverage. This is contrasted by other investigations that take advantage of large samples of available time series with sufficient observations (Berghuijs et al., 2017; Do et al., 2017), thereby providing a richer spatial picture of changes in water availability.

As for regional studies, there is a large heterogeneity between the individual global assessments, including a wide range of research questions, different spatial sampling schemes, and different time periods. Some studies are dedicated to investigating mean flows (Alkama et al., 2011; Dai et al., 2009; Dai & Trenberth, 2002; Jaramillo & Destouni, 2015; Labat et al., 2004; Milliman et al., 2008; Milly et al., 2005), while others focus on floods (Berghuijs et al., 2017; Do et al., 2017; Kundzewicz et al., 2005; Svensson et al., 2005) or low-flow indicators (Svensson et al., 2005). In summary, the heterogeneity of past global-scale assessments makes it difficult to draw generalized conclusions on observable changes of streamflow around the world.

This study updates previous assessments of worldwide changes in streamflow, using a database with unprecedented spatial coverage of streamflow observations and using indicators of low, mean, and high flows. To account for regional differences in data availability, trends are analyzed for three overlapping 40-year periods from 1951 to 2010, maximizing the spatiotemporal coverage of the investigation. Finally, the significance of the observed trends is established at the subcontinental scale.

### 2. Data

Streamflow observations are taken from the Global Streamflow Indices and Meta data archive (GSIM; Do et al., 2018b; Gudmundsson et al., 2018b), which is available in the public domain (Do et al., 2018a; Gudmundsson et al., 2018a) and holds information from more than 30,000 gauging stations. Annual time series information is available through indices computed from daily values that represent a wide range of flow properties at monthly, seasonal, and yearly resolution. Here the following indices are considered:

- 1. Low flows are represented through time series of the annual minimum (MIN) and the annual 10th percentile (P10).
- 2. Average flow conditions are characterized using the annual mean (MEAN) and the annual 50th percentile/median (P50).
- 3. High flows are represented through time series of the annual maximum (MAX) and the annual 90th percentile (P90).

Daily time series used to compute the GSIM indices underwent a formal quality assessment (Gudmundsson et al., 2018b). The assessment utilized quality flags from individual data providers and automated screening methods that flag implausible values. Only daily records that passed this assessment were used for index calculation. Because the extremal indices (MIN, P10, P90, and MAX) are sensitive to data availability, years with less than 350 valid daily observations were set to missing for each station, as recommended by ECA, & D Project Team, and K. Royal Netherlands Meteorological Institute (2013; hereafter ECA&D13).

Note also that GSIM combines information from all gauging stations from the contributing data bases. Consequently, both near-natural and regulated catchments are included (Do et al., 2018b; Gudmundsson et al., 2018b). In this study no attempt is made to distinguish between these cases. Instead, trends in the complete observational record are documented, as changes in atmospheric boundary conditions and human water management might both trigger changes in streamflow.



SREX regions

**Figure 1.** Subcontinental regions defined by the Special Report On Extremes (SREX; Seneviratne et al., 2012). Top: world map of all regions, where regions with more than 50 stations with sufficient data in at least one of the three considered 40-year periods are highlighted in red. Bottom: number of stations with sufficient data for each period and each SREX region. The red line indicates the 50-station threshold.

SAS: South Asia

Given the complex nature of in situ observations entering the GSIM archive, spatial and temporal coverage of the considered streamflow time series varies substantially around the globe. Therefore, and because trends can be influenced by decadal variability, the following 40-year periods were analyzed: 1951–1990, 1961–2000, and 1971–2010. Based on previously suggested data availability criteria (ECA&D13) for trend analysis, only stations where at least 70% of the years are available were considered. This criterion was applied to each of the 40-year periods separately. As a result, the spatial coverage differs across the periods.

The significance of trends is evaluated at the subcontinental scale by grouping stations into 26 regions that were designed for analyzing regional climate change and are defined in the Special Report on Extremes of the Intergovernmental Panel on Climate Change (Seneviratne et al., 2012; later referred to as SREX regions). Figure 1 shows the SREX regions alongside the number of stations that fulfill the data availability criteria for each region and each 40-year period. Only regions and periods with at least 50 stations were considered for subcontinental-scale assessment.

# 3. Trend Estimation

EAS: East Asia

Following previous studies (Stahl et al., 2010, 2012), trends at individual stations were computed using the robust Sen-Theil slope estimator (Sen, 1968). To make trend estimates from catchments with different sizes



and from different climates comparable, they are expressed in units of percent change per decade (i.e., 10 years; Stahl et al., 2012), such that

$$T_s = \frac{\tau_s \times 10 \text{ years}}{\bar{x}_s} \times 100 \tag{1}$$

where  $T_s$  is the trend at location *s* in units of percentage change per decade,  $\tau_s$  is the Sen-Theil slope estimator, and  $\overline{x}_s$  is the mean of the index time series.

To be able to detect changes at the level of the SREX regions, a resampling method is proposed that accounts for within-region spatial dependence (Burn & Elnur, 2002; Douglas et al., 2000; Wilks, 2011).

The regional trend test is as follows:

- 1. For a given region, compute the regional trend,  $\overline{T_s}$  defined as the average of all  $T_s$  in that region.
- 2. Repeat 2,000 times:
- 2.1 Resample with replacement the year order of all data within the region while maintaining the spatial dependence within individual years, following the procedure described in Burn and Elnur (2002).
- 2.2 Compute at each location  $T_s^*$ , the trend expressed in percent change per decade of the resampled time series.
- 2.3 Compute the regional trend,  $\overline{T}_s^*$ , as the regional mean of  $T_s^*$ .
- 3. Estimate *p*, that is, the probability of  $\overline{T_s}$  on the distribution of  $\overline{T_s}^*$  (the bootstrap distribution) as the fraction of  $\overline{T_s} < \overline{T_s}^*$ .

Significance of regional trends is reported at the p < 0.01 and p < 0.1 level for negative trends and at the p > 0.9 and p > 0.99 level for positive trends.

Note that this procedure is related to previously suggested closed-form (Helsel & Frans, 2006) and resampling-based (Douglas et al., 2000) regional adaptations of the Mann-Kendal trend test. However, the method introduced here does not require the additional step of computing the Mann-Kendall statistic. Instead, it operates on the variable of interest, the regional trend ( $\overline{T_s}$ ). We note that regional testing procedures have the inherent limitation that they cannot consider subregional variability, with the potential for groups of stations with positive and negative trends to mask each other out.

#### 4. Results and Discussion

#### 4.1. Overview

Figure 2 maps the trend magnitude of the time series indices for each period. Visual inspection of the results highlights that streamflow is not changing uniformly around the world and that the considered period can have significant effects on both the sign and magnitude of the trend.

To better understand the nature of the observed trends, Figure 3 shows the regional trends, which often point in the same direction across all indices. In the following, these regional changes will be summarized and discussed in the context of selected observational studies. In addition, the observed change patterns will be put into the context of model projections of water availability (precipitation minus evapotranspiration; Greve et al., 2018) and runoff (Haddeland et al., 2014). Note that the aforementioned studies are based on different model ensembles, each having their distinct characteristics. Greve et al. (2018) is based on the CIMIP5 ensemble (Taylor et al., 2011) that includes a large sample of global climate models but does not account for human water management and land cover change. Conversely, Haddeland et al. (2014) is based on the Inter-Sectoral Impact Model Intercomparison Project (ISIMIP) Fast Track ensemble (Warszawski et al., 2014) of global hydrological models driven with selected global climate models that accounts for both climate change and on-ground human activities. Finally, it is noted that ocean-atmosphere oscillations can be an important influence on decadal streamflow variability that have been studied elsewhere in great detail, including, for example, global (Wanders & Wada, 2015; Ward et al., 2010), North America (Burn, 2008; Tootle et al., 2005; Tootle & Piechota, 2006), Europe (Bouwer et al., 2006, 2008; Kingston et al., 2012), and Australia (Kuhnel et al., 1990; Verdon et al., 2004) assessments.

#### 4.2. North America

North America has the highest number of stations of all the continents considered. In West North America (WNA) there is no consistent change pattern. Increasing regional average low flows (MIN



**Figure 2.** Trends in annual indicators of mean and extreme streamflow. Columns represent three 40-year periods. Rows represent the annual minimum (MIN), the annual 10th percentile (P10), the annual 50th percentile (P50), the annual MEAN (MEAN), the annual 90th percentile (P90), and the annual maximum (MAX). SREX regions with at least 50 stations with sufficient data are highlighted. See supporting information for high-resolution maps.

and P10) are detected for the 1951–1990 and 1961–2000 periods, whereas decreasing mean annual flows (MEAN) and high flows (P90) are found in the 1971–2010 period. In contrast, streamflow has increased significantly in Central North America (CNA) throughout the first two periods (1951–1990 and 1961–2000) across all indices, except for the annual maximum flow (MAX). This wetting tendency weakens in 1971–2010, where only MIN and P10 show a significantly increasing regional trend. There has been an increase in streamflow in East North America (ENA) over the first two periods (1951–1990 and 1961–2000), which is most pronounced for low flows and mean flows but less pronounced for high flows. In 1971–2010, the change pattern reversed, with a significantly declining regional trend for all indices except P90.

Overall, the results confirm previous assessments that focus on observations prior to the year 2000 in the United States (Douglas et al., 2000; Lettenmaier et al., 1994; Lins & Slack, 1999; McCabe & Wolock, 2002). These studies emphasize the tendency for increasing low and mean flows throughout the region, which is particularly pronounced in the central north of the U.S. There is also agreement in the lack of observed annual maximum trends, although only a few studies have focused on the period after 2000 (Hodgkins et al., 2017; Mallakpour & Villarini, 2015).

Based on simulations of the CMIP5 ensemble, Greve et al. (2018) report a clear tendency toward wetter conditions in WNA and ENA with no clear change pattern in CNA. Conversely, Haddeland et al. (2014) report a





Figure 3. Regional trends for SREX regions with at least 50 stations in one of the considered periods. Regional trends are computed for all indicators of mean and extreme streamflow. Significance of the regional trend is reported as the probability of the observed trend on the bootstrap distribution. Regional trends are only provided for periods with at least 50 stations.

tendency for decreasing water availability, especially in the south of North America, which is triggered by human water and land management as simulated in the considered ISIMIP model ensemble. None of these are directly comparable with the observed change patterns, which exhibit shifting signs in both WNA and ENA throughout the study period.

#### 4.3. South America

In South America, data availability increases throughout the study period. In the Amazon region (AMZ) only the 1971–2010 period has more than 50 stations available. In this period, the median (P50) and high flows (P90 and MAX) show a significant increasing regional trend. In North-East Brazil (NEB) all indices except P10 show a significant increasing regional trend in the 1951–1990 period. This pattern reverses thereafter, and all indices exhibit a negative regional trend for 1961–2000, although not all are significant. In the 1971–2010 period all indices show a significant declining regional trend. Southeastern South America (SSA) has wetting trends in all indices in the first period. In the second period all indices except P90 also show significant increasing trends, but this increasing tendency comes to an abrupt stop in the latter period, where no significant trends are found in most of the indices.

Analyzing streamflow observations in the second half of the twentieth century, Marengo et al. (1998) did find mostly positive trends in the region comprising parts of AMZ and NEB. This is partly consistent with the presented results but does not report the reversing trend pattern in NEB occurring between the first and second periods. For rivers draining to the south Atlantic, Pasquini and Depetris (2007) report complex spatiotemporal trend patterns using observations up to the early 2000s. For a region similar to SSA, Genta et al. (1998) report increasing discharge trends for the second half of the twentieth century with a tendency to level off, which is in agreement with the presented results.

Overall, climate models from the CMIP5 ensemble suggest that both AMZ and NEB have a tendency for becoming drier with increasing global mean temperatures, while SSA is likely to become wetter (Greve et al., 2018). However, an alternative ensemble suggests that SSA might also become increasingly drier in a warmer climate (Haddeland et al., 2014). On-ground water and land management is estimated to only have a limited impact on freshwater resources in most parts of South America (Haddeland et al., 2014).

#### 4.4. Europe

Europe is among the best monitored regions with respect to streamflow around the world. In North Europe (NEU), only P90 exhibited a weak increasing regional trend in the 1951–1990 period. In 1961–2000 all indices show a weak incline, while only MEAN, P90, and MAX are significant. In the 1971–2010 period all indices showed a weak increasing trend, with significant MIN, P10, MEAN, and MAX suggesting a slight upward shift of the annual daily streamflow distribution. In Central Europe (CEU), there is a significant upward regional trend in MIN, P10, and P50 in the first period (1951–1990). In the subsequent period (1961–2000) almost no changes occur except for MIN, which shows a weak positive regional trend. In the last period (1971–2010) only MAX shows a weak significant regional trend in CEU. The South Europe/Mediterranean (MED) region (note that all considered stations are in Europe) shows the strongest and the most consistent pattern of the entire study. Here all indices show strong and significant declining regional trends throughout all considered time periods, highlighting an overall reduction of freshwater availability in this region.

Studies focusing on trends in flood frequency (Hodgkins et al., 2017) and drought indicators (Hisdal et al., 2001) report that there is little evidence for changes in these quantities in Europe. However, several studies have documented the tendency for drying in the South of Europe and increasingly wet conditions in the north (Stahl et al., 2010, 2012). Through linking this observational pattern to historical climate model simulations, the observed trends in pan-European freshwater availability have been attributed to anthropogenic climate change (Gudmundsson et al., 2017).

Future climate projections indicate a continuation of the observed trend pattern in Europe with increasingly dry conditions in MED, wetting conditions in NEU, and little change in CEU (Greve et al., 2018; Haddeland et al., 2014). In addition, Haddeland et al. (2014) indicate that human water and land management may also contribute to declining streamflow values in southern Europe, which might have amplified the observed strong negative trend throughout all aspects of the flow distribution.

#### 4.5. Asia

Spatiotemporal data availability is variable over the Asian continent. In North Asia (NAS), all indices except MAX show a significant increasing regional trend for the 1951–1990 and 1961–2000 periods. In the last period (1971–2010) there are less than 50 stations in this large region. In South Asia (SAS) only the 1971–2010 period passes the data availability criteria, having a significantly declining regional trend in MEAN, P90, and MAX. Relatively better data coverage is found in East Asia (EAS), with sufficient data to cover the last two time periods (note that most stations are in Japan). For 1961–2000, MIN, P10, MEAN, and MAX show significant increasing regional trends. For 1971–2010, the regional trends of MIN and P10 are increasing significantly.

Several previous studies have documented increasing streamflow trends in the north of the Asian continent (Adam & Lettenmaier, 2008; MacDonald et al., 2007; McClelland et al., 2006; Tananaev et al., 2016), which appears to persist past the year 2000 and is also visible in hydrological extremes (Tananaev et al., 2016). A regional study in the Indian subcontinent confirms the tendency toward decreased water availability and attributes it to anthropogenic climate change (Mondal & Mujumdar, 2012).

Global climate models project that water availability will increase in NAS as a consequence of global warming (Greve et al., 2018). This is consistent with the observational results of the present study, and only limited impacts of human management on water resources is expected (Haddeland et al., 2014). In SAS climate models indicate that global warming will increase water availability (Greve et al., 2018), contrasting our observational findings. However, the simulations assessed by Haddeland et al. (2014) suggest that human land management is reducing runoff in the Indian subcontinent offering a possible explanation for the observed signal. Also in EAS, climate models project increasing water availability in an warming climate (Greve et al., 2018) and impacts of water and land management are only moderate (Haddeland et al., 2014).

#### 4.6. Africa

Of the entire continent of Africa, only Southern Africa (SAF) has more than 50 stations fulfilling the data availability requirements. In the 1951–1990 period, low-flow indices (MIN and P10) and the annual median (P50) show significant decreasing regional trends. In 1961–2000 this weak drying pattern is reinforced, and all indices except P90 show significant negative trends, pointing at an overall decrease throughout the runoff distribution. In the 1971–2010 period, however, this pattern weakens and only MIN shows a significant decline.

A comprehensive assessment of trends in a region similar to SAF found more decreasing than increasing trends (Fanta et al., 2001), which is consistent with the findings of the present study. Overall, global climate models suggest that increasing global mean temperatures are associated with drying conditions in SAF (Greve et al., 2018), which might even be intensified through reduced flow rates triggered by human water and land management (Haddeland et al., 2014).

#### 4.7. Oceania

In Oceania, only South Australia/New Zealand (SAU) has sufficient data coverage to warrant analysis. Data availability is not sufficient in the period 1951–1990. In 1961–2000, negligible change is observed, except weak inclination in MIN. However, the last period (1971–2010) shows strong and significant negative regional trends of all indices considered, that is, a strong and significant reduction of the entire flow distribution.

Previous assessments of changes in streamflow (Petrone et al., 2010; Zhang et al., 2016) and annual maximum floods (Ishak et al., 2013) have reported declining trends in southern Australia. In southeastern Australia, the first decade of the 21st century was particularly dry, sometimes referred to as the millennium drought (Kiem et al., 2016; Low et al., 2015; Zhang et al., 2016).

On average, the climate models of the CMIP5 ensemble indicate only a weak change in water availability in SAU with increasing global mean temperatures (Greve et al., 2018). However, other simulations suggest that both anthropogenic climate change and human water use may trigger a significant reduction of runoff in south eastern Australia (Haddeland et al., 2014), which is consistent with the observed changes.

## 5. Summary and Conclusions

To date, there has been low confidence and a lack of consistent evidence regarding sign and magnitude of trends in global river discharge during the twentieth century (Hartmann et al., 2013). Therefore, this study presents a comprehensive update of global-scale changes in indicators of mean and extreme streamflow taken from the GSIM archive (Do et al., 2018b; Gudmundsson et al., 2018b). To enable this global overview across all indicators, the focus of the analyses has been the significance and sign of change at the subcontinental scale, contrasting the common approach to solely report trend magnitudes for individual stations. In

contrast to regional studies tailored to specific indices and with varying methods, a key benefit of this study was the opportunity to consider multiple regions and multiple indices with a consistent method. The subsequent analysis highlights that streamflow trends have complex spatial patterns, preventing simple generalizations of regional changes to the global scale.

A striking result is that in most cases the sign of regional trends is consistent across all indices. This implies that the entire flow distribution is changing upward or downward in the respective regions, indicating generally wetter or drier conditions. In other words, increasing low flows are in most cases associated with increasing high flows (and vice versa), contradicting the common notion that flood and drought risk may increase simultaneously. Another feature of the results is that for some regions (West North America, East North America, and North-East Brazil) the sign of the trends has varied with respect to the considered period, suggesting low-frequency variability in the baseline climate signal and that care is needed in the interpretation of the associated change patterns.

Among all considered regions, South Europe/Mediterranean had the strongest signal with consistent negative trends in all indices throughout all considered time periods. Other regions with predominantly negative trends include Southern Africa, South Australia/New Zealand, and potentially South Asia. In addition, Northeastern Brazil experienced drying conditions for the last two time periods but had a consistent wetting trend for the first period. Consistent wetting trends were observed in Central North America, Southeastern South America, North Europe, and North Asia, although the trend weakens for the last period in Central North America and Southeast South America. Overall, these wetting trends are not equally visible in all regions and throughout all indices.

While the number of gauges used in this study is unprecedented, the conclusions in regions with less data are constrained (e.g., Asia) or muted (e.g., Africa), and further gains in data gathering would substantially improve confidence (Do et al., 2018b). Consequently, spatiotemporal coverage of the observations remains a limiting factor. Likewise, both the potentially uneven temporal distribution of available data in individual time series and regional differences in spatial coverage are impacting the results. Finally, the focus on regional trends can mask subregional features. Nevertheless, the presented results provide for an unprecedented view on streamflow trends around the world.

While this study has sought to interpret observed changes in the context of future climate projections (Greve et al., 2018) and model estimates of the impacts of human water and land management (Haddeland et al., 2014), it does not allow for a conclusive attribution of the observed changes to either of these factors. To this end, formal detection and attribution methods (Bindoff et al., 2013; Gudmundsson et al., 2017) are needed, which would allow for systematic testing of the hypothesis that both anthropogenic climate change and human water and land management are impacting renewable freshwater resources and hydrological extremes at the global scale. As end of the century projections of global water resources and hydrological extremes increase in number and sophistication, an appreciation for trends in observed indicators of mean and extreme streamflow provides a stronger basis for understanding future changes.

#### Acknowledgments

All streamflow indices considered are taken from the GSIM archive and are freely available from https://doi.org/ 10.1594/PANGAEA.887470 (Gudmundsson et al., 2018a). Mr. Hong Xuan Do receives financial support from the Australia Award Scholarship (AAS) and D. R. Stranks Traveling Fellowship. This work is partially funded by the ERC DROUGHT-HEAT project (contract 617518).

#### References

Adam, J. C., & Lettenmaier, D. P. (2008). Application of new precipitation and reconstructed streamflow products to streamflow trend attribution in northern Eurasia. *Journal of Climate*, 21(8), 1807–1828.

Alkama, R., Decharme, B., Douville, H., & Ribes, A. (2011). Trends in global and basin-scale runoff over the late twentieth century: Methodological issues and sources of uncertainty. *Journal of Climate*, 24(12), 3000–3014.

Berghuijs, W. R., Aalbers, E. E., Larsen, J. R., Trancoso, R., & Woods, R. A. (2017). Recent changes in extreme floods across multiple continents. *Environmental Research Letters*, 12(11), 114,035–114,035.

Bindoff, N. L., Stott, P. A., AchutaRao, K. M., Allen, M. R., Gillett, N., Gutzler, D., et al. (2013). Detection and attribution of climate change: From global to regional. In T. F. Stocker, D. Qin, G. K. Plattner, M. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 867–952). Cambridge, UK: Cambridge University Press.

Blöschl, G., Hall, J., Parajka, J., Perdigão, R. A. P., Merz, B., Arheimer, B., et al. (2017). Changing climate shifts timing of European floods. Science, 357(6351), 588–590. https://doi.org/10.1126/science.aan2506

Bouwer, L. M., Vermaat, J. E., & Aerts, J. C. J. H. (2006). Winter atmospheric circulation and river discharge in northwest Europe. Geophysical Research Letters, 33, L06403. https://doi.org/10.1029/2005GL025548

Bouwer, L. M., Vermaat, J. E., & Aerts, J. C. J. H. (2008). Regional sensitivities of mean and peak river discharge to climate variability in Europe. *Journal of Geophysical Research*, 113, D19103. https://doi.org/10.1029/2008JD010301

Burn, D. H. (2008). Climatic influences on streamflow timing in the headwaters of the Mackenzie River Basin. *Journal of Hydrology*, 352(1–2), 225–238.

Burn, D. H., & Elnur, M. A. H. (2002). Detection of hydrologic trends and variability. *Journal of Hydrology*, 255(1–4), 107–122. Dai, A., Qian, T., Trenberth, K. E., & Milliman, J. D. (2009). Changes in continental freshwater discharge from 1948 to 2004. *Journal of Climate*, 22(10), 2773–2792.

Dai, A., & Trenberth, K. E. (2002). Estimates of freshwater discharge from continents: Latitudinal and seasonal variations. Journal of Hydrometeorology, 3(6), 660–687.

Do, H. X., Gudmundsson, L., Leonard, M., & Westra, S. (2018a). The global streamflow indices and metadata archive—Part 1: Station catalog and catchment boundary, edited, PANGAEA.

Do, H. X., Gudmundsson, L., Leonard, M., & Westra, S. (2018b). The Global Streamflow Indices and Metadata archive (GSIM)—Part 1: The production of a daily streamflow archive and metadata. *Earth System Science Data*, 10(2), 765–785.

Do, H. X., Westra, S., & Leonard, M. (2017). A global-scale investigation of trends in annual maximum streamflow. Journal of Hydrology, 552, 28–43.

Douglas, E. M., Vogel, R. M., & Kroll, C. N. (2000). Trends in floods and low flows in the United States: Impact of spatial correlation. *Journal of Hydrology*, 240(1–2), 90–105.

ECA & D Project Team, and K. Royal Netherlands Meteorological Institute (2013). Algorithm Theoretical Basis Document (ATBD). Rep. Retrieved from https://www.ecad.eu/documents/atbd.pdf

Fanta, B., Zaake, B. T., & Kachroo, R. K. (2001). A study of variability of annual river flow of the southern African region. Hydrological Sciences Journal, 46(4), 513–524.

Fekete, B. M., Looser, U., Pietroniro, A., & Robarts, R. D. (2012). Rationale for monitoring discharge on the ground. Journal of Hydrometeorology, 13(6), 1977–1986.

Fekete, B. M., Robarts, R. D., Kumagai, M., Nachtnebel, H.-P., Odada, E., & Zhulidov, A. V. (2015). Time for in situ renaissance. *Science*, 349(6249), 685–686.

Genta, J. L., Perez-Iribarren, G., & Mechoso, C. R. (1998). A recent increasing trend in the streamflow of rivers in southeastern South America. Journal of Climate, 11(11), 2858–2862.

Greve, P., Gudmundsson, L., & Seneviratne, S. I. (2018). Regional scaling of annual mean precipitation and water availability with global temperature change. *Earth System Dynamics*, 9(1), 227–240.

Gudmundsson, L., Do, H. X., Leonard, M., & Westra, S. (2018a). The Global Streamflow Indices and Metadata archive (GSIM)—Part 2: Time series indices and homogeneity assessment, edited, PANGAEA.

Gudmundsson, L., Do, H. X., Leonard, M., & Westra, S. (2018b). The Global Streamflow Indices and Metadata archive (GSIM)—Part 2: Quality control, time-series indices and homogeneity assessment. *Earth System Science Data*, 10(2), 787–804.

Gudmundsson, L., Seneviratne, S. I., & Zhang, X. (2017). Anthropogenic climate change detected in European renewable freshwater resources. *Nature Climate Change*, 7, 813–816.

Haddeland, I., Heinke, J., Biemans, H., Eisner, S., Flörke, M., Hanasaki, N., et al. (2014). Global water resources affected by human interventions and climate change. Proceedings of the National Academy of Sciences of the United States of America, 111(9), 3251–3256. https://doi.org/10.1073/pnas.1222475110

Hannaford, J., Buys, G., Stahl, K., & Tallaksen, L. M. (2013). The influence of decadal-scale variability on trends in long European streamflow records. *Hydrology and Earth System Sciences*, 17(7), 2717–2733.

Hannah, D. M., Demuth, S., van Lanen, H. A. J., Looser, U., Prudhomme, C., Rees, G., et al. (2011). Large-scale river flow archives: Importance, current status and future needs. *Hydrological Processes*, 25(7), 1191–1200.

 Hartmann, D. L., Klein Tank, A. M. G., Rusicucci, M., Alexander, L. V., Broenniman, B., Charabi, Y., et al. (2013). Observations: Atmosphere and surface. In T. F. Stocker, D. Qin, G. K. Plattner, M. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), *Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 159–254). Cambridge, UK: Cambridge University Press.

Helsel, D. R., & Frans, L. M. (2006). Regional Kendall test for trend. Environmental Science & Technology, 40(13), 4066–4073.

Hirabayashi, Y., Mahendran, R., Koirala, S., Konoshima, L., Yamazaki, D., Watanabe, S., et al. (2013). Global flood risk under climate change. *Nature Climate Change*. https://doi.org/10.1038/nclimate1911

Hisdal, H., Stahl, K., Tallaksen, L. M., & Demuth, S. (2001). Have streamflow droughts in Europe become more severe or frequent? International Journal of Climatology, 21, 317–333.

Hodgkins, G. A., Whitfield, P. H., Burn, D. H., Hannaford, J., Renard, B., Stahl, K., et al. (2017). Climate-driven variability in the occurrence of major floods across North America and Europe. *Journal of Hydrology*, 552, 704–717. https://doi.org/10.1016/j.jhydrol.2017.07.027

Ishak, E. H., Rahman, A., Westra, S., Sharma, A., & Kuczera, G. (2013). Evaluating the non-stationarity of Australian annual maximum flood. *Journal of Hydrology*, *49*, 134–145.

Jaramillo, F., & Destouni, G. (2015). Local flow regulation and irrigation raise global human water consumption and footprint. Science, 350(6265), 1248–1251.

Kiem, A. S., Johnson, F., Westra, S., van Dijk, A., Evans, J. P., O'Donnell, A., et al. (2016). Natural hazards in Australia: Droughts. Climatic Change, 139(1), 37–54. https://doi.org/10.1007/s10584-016-1798-7

Kingston, D. G., Fleig, A. K., Tallaksen, L. M., & Hannah, D. M. (2012). Ocean-atmosphere forcing of summer streamflow drought in Great Britain. Journal of Hydrometeorology, 14(1), 331–344.

Kuhnel, I., McMahon, T. A., Finlayson, B. L., Haines, A., Whetton, P. H., & Gibson, T. T. (1990). Climatic influences on streamflow variability: A comparison between southeastern Australia and southeastern United States of America. *Water Resources Research*, 26(10), 2483–2496.

Kundzewicz, Z. W., Graczyk, D., Maurer, T., Pinskwar, I., Radziejewski, M., Svensson, C., & Szwed, M. g. (2005). Trend detection in river flow series: 1. Annual maximum flow. *Hydrological Sciences-Journal-des Sciences Hydrologiques*, 50(5), 797–810.

Labat, D., Goddéris, Y., Probst, J. L., & Guyot, J. L. (2004). Evidence for global runoff increase related to climate warming. *Advances in Water Resources*, *27*(6), 631–642.

Lettenmaier, D. P., Wood, E. F., & Wallis, J. R. (1994). Hydro-climatological trends in the continental United States, 1948-88. Journal of Climate, 7(4), 586-607.

Lins, H. F., & Slack, J. R. (1999). Streamflow trends in the United States. Geophysical Research Letters, 26(2), 227–230. https://doi.org/ 10.1029/1998GL900291

Low, K. G., Grant, S. B., Hamilton, A. J., Gan, K., Saphores, J.-D., Arora, M., & Feldman, D. L. (2015). Fighting drought with innovation: Melbourne's response to the Millennium Drought in Southeast Australia. *Wiley Interdisciplinary Reviews Water*, 2(4), 315–328.

MacDonald, G. M., Kremenetski, K. V., Smith, L. C., & Hidalgo, H. G. (2007). Recent Eurasian river discharge to the Arctic Ocean in the context of longer-term dendrohydrological records. Journal of Geophysical Research, 112, G04S50. https://doi.org/10.1029/2006JG000333



Mallakpour, I., & Villarini, G. (2015). The changing nature of flooding across the central United States. Nature Climate Change, 5(3), 250–254.

Marengo, J. A., Tomasella, J., & Uvo, C. R. (1998). Trends in streamflow and rainfall in tropical South America: Amazonia, eastern Brazil, and northwestern Peru. Journal of Geophysical Research, 103(D2), 1775–1783. https://doi.org/10.1029/97JD02551

McCabe, G. J., & Wolock, D. M. (2002). A step increase in streamflow in the conterminous United States. *Geophysical Research Letters*, 29(24), 2185. https://doi.org/10.1029/2002GL015999

- McClelland, J. W., Déry, S. J., Peterson, B. J., Holmes, R. M., & Wood, E. F. (2006). A pan-Arctic evaluation of changes in river discharge during the latter half of the 20th century. *Geophysical Research Letters*, 33, L06715. https://doi.org/10.1029/2006GL025753
- Milliman, J. D., Farnsworth, K. L., Jones, P. D., Xu, K. H., & Smith, L. C. (2008). Climatic and anthropogenic factors affecting river discharge to the global ocean, 1951--2000. Global and Planetary Change, 62(3-4), 187-194.
- Milly, P. C. D., Dunne, K. A., & Vecchia, A. V. (2005). Global pattern of trends in streamflow and water availability in a changing climate. *Nature*, 438(7066), 347–350.
- Mondal, A., & Mujumdar, P. P. (2012). On the basin-scale detection and attribution of human-induced climate change in monsoon precipitation and streamflow. *Water Resources Research*, 48, W10520. https://doi.org/10.1029/2011WR011468

Pasquini, A. I., & Depetris, P. J. (2007). Discharge trends and flow dynamics of South American rivers draining the southern Atlantic seaboard: An overview. Journal of Hydrology, 333(2), 385–399.

Petrone, K. C., Hughes, J. D., van Niel, T. G., & Silberstein, R. P. (2010). Streamflow decline in southwestern Australia, 1950-2008. Geophysical Research Letters, 37, L11401. https://doi.org/10.1029/2010GL043102

Prudhomme, C., Giuntoli, I., Robinson, E. L., Clark, D. B., Arnell, N. W., Dankers, R., et al. (2014). Hydrological droughts in the 21st century, hotspots and uncertainties from a global multimodel ensemble experiment. *Proceedings of the National Academy of Sciences of the United States of America*, 111(9), 3262–3267. https://doi.org/10.1073/pnas.1222473110

Schewe, J., Heinke, J., Gerten, D., Haddeland, I., Arnell, N. W., Clark, D. B., et al. (2014). Multimodel assessment of water scarcity under climate change. Proceedings of the National Academy of Sciences of the United States of America, 111(9), 3245–3250. https://doi.org/ 10.1073/pnas.1222460110

- Sen, P. K. (1968). Estimates of the regression coefficient based on Kendall's Tau. Journal of the American Statistical Association, 63(324), 1379–1389.
- Seneviratne, S. I., Nicholls, N., Easterling, D., Goodess, C. M., Kanae, S., Kossin, J., et al. (2012). Changes in climate extremes and their impacts on the natural physical environment. In C. B. Field, V. Barros, T. F. Stocker, D. Qin, D. J. Dokken, K. L. Ebi et al. (Eds.), Managing the risks of extreme events and disasters to advance climate change adaptation, A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change (IPCC) (pp. 109–230). Cambridge, UK and New York: Cambridge University Press. Retrieved from https://www.ipcc.ch/site/assets/uploads/2018/03/SREX-Chap3\_FINAL-1.pdf
- Stahl, K., Hisdal, H., Hannaford, J., Tallaksen, L. M., van Lanen, H. A. J., Sauquet, E., et al. (2010). Streamflow trends in Europe: Evidence from a dataset of near-natural catchments. *Hydrology and Earth System Sciences*, 14(12), 2367–2382.

Stahl, K., Tallaksen, L. M., Hannaford, J., & van Lanen, H. A. J. (2012). Filling the white space on maps of European runoff trends: Estimates from a multi-model ensemble. *Hydrology and Earth System Sciences*, *16*(7), 2035–2047.

Svensson, C., Kundzewicz, W. Z., & Maurer, T. (2005). Trend detection in river flow series: 2. Flood and low-flow index series. Hydrological Sciences-Journal-des Sciences Hydrologiques, 50(5), 811–824.

Tananaev, N. I., Makarieva, O. M., & Lebedeva, L. S. (2016). Trends in annual and extreme flows in the Lena River basin, northern Eurasia. Geophysical Research Letters, 43, 10,764–10,772. https://doi.org/10.1002/2016GL070796

Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2011). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498.

Tootle, G. A., & Piechota, T. C. (2006). Relationships between Pacific and Atlantic Ocean sea surface temperatures and U.S. streamflow variability. *Water Resources Research*, 42, W07411. https://doi.org/10.1029/2005WR004184

Tootle, G. A., Piechota, T. C., & Singh, A. (2005). Coupled oceanic-atmospheric variability and U.S. streamflow. *Water Resources Research*, 41, W12408. https://doi.org/10.1029/2005WR004381

Verdon, D. C., Wyatt, A. M., Kiem, A. S., & Franks, S. W. (2004). Multidecadal variability of rainfall and streamflow: Eastern Australia. Water Resources Research, 40, W10201. https://doi.org/10.1029/2004WR003234

Wanders, N., & Wada, Y. (2015). Decadal predictability of river discharge with climate oscillations over the 20th and early 21st century. Geophysical Research Letters, 42, 10,689–10,695. https://doi.org/10.1002/2015GL066929

Ward, P. J., Beets, W., Bouwer, L. M., Aerts, J. C. J. H., & Renssen, H. (2010). Sensitivity of river discharge to ENSO. Geophysical Research Letters, 37, L12402. https://doi.org/10.1029/2010GL043215

 Warszawski, L., Frieler, K., Huber, V., Piontek, F., Serdeczny, O., & Schewe, J. (2014). The Inter-Sectoral Impact Model Intercomparison Project (ISI-MIP): Project framework. Proceedings of the National Academy of Sciences of the United States of America, 111(9), 3228–3232.
Wilks, D. S. (2011). Statistical methods in the atmospheric sciences (p. 676). Oxford, Amsterdam, and Waltham, MA: Academic Press.

Zhang, X. S., Amirthanathan, G. E., Bari, M. A., Laugesen, R. M., Shin, D., Kent, D. M., et al. (2016). How streamflow has changed across Australia since the 1950s: Evidence from the network of hydrologic reference stations. *Hydrology and Earth System Sciences*, 20(9), 3947–3965.